Author’s response to reviews

Title: Potent inhibition of rhabdoid tumor cells by combination of flavopiridol and 4OH-Tamoxifen

Authors:

Velasco Cimica (v CIMICA@NOTES.CC.SUNYSB.EDU)
Melissa E Smith (melissa.smith@phd.einstein.yu.edu)
Zhikai Zhnag (zzhang@aecom.yu.edu)
Deepti Mathur (dm322@cornell.edu)
Sridhar Mani (sridhar.mani@einstein.yu.edu)
Ganjam V Kalpana (ganjam.kalpana@einstein.yu.edu)

Version: 5 Date: 27 August 2010

Author’s response to reviews:

Dear Sir or Madam,

We are submitting a revised version of the manuscript text, which now includes a Competing Interests section as the editors requested. The Competing Interests section states that “The authors declare that they have no competing interests,” and can be found on page 21 of the revised manuscript. Thank you for your time in reviewing this manuscript for publication in BMC Cancer.

Sincerely,

Ganjam V. Kalpana
Professor